Author’s response to reviews

Title: MHealth and Perceived Quality of Care Delivery: A Conceptual Model and Empirical Validation

Authors:

Yvonne O'Connor (y.oconnor@ucc.ie)

Pavel Andreev (Andreev@telfer.uottawa.ca)

Philip O'Reilly (Philip.ORelly@ucc.ie)

Version: 3 Date: 07 Feb 2020

Author’s response to reviews:

Please ensure that Philip O’ Reilly has been affiliated to an institution in the title page. - This is complete

--Please amend the 'Error! Reference source not found' within the Results section. - This is complete

-- Please acknowledge anyone who contributed towards the article who does not meet the criteria for authorship including anyone who provided professional writing services or materials. Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

If there are no acknowledgements, please state “Not applicable”. - This is completed

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. - This is completed.